Author's response to reviews

Title: Relationship between ALDH2 Genotype and In-stent Restenosis in Chinese Han Patients after Percutaneous Coronary Intervention

Authors:

Lizhi Lv (provlizhi@126.com)
Weijie Ye (463111153@qq.com)
Peiyuan Song (song_ng_peiyuan@126.com)
Yubin Chen (470643668@qq.com)
Jing Yang (fccyangj3@zzu.edu.cn)
Congmin Zhang (fcczhangcm@zzu.edu.cn)
Xiaoping Chen (chenxp74@hotmail.com)
Fanyan Luo (drlfy@163.com)

Version: 1 Date: 09 Mar 2019

Author's response to reviews:

Dear Editor:

We would like to thank the editor for giving us a chance to resubmit our paper entitled "Relationship between ALDH2 Genotype and In-stent Restenosis in Chinese Han Patients after Percutaneous Coronary Intervention" (BCAR-D-18-00746). We modified the manuscript according to your suggestions.

1. FORMATTING CHANGES: we upload a clean version of revised submission.

2. DECLARATIONS SECTION: We completed the Declarations section.
For the 'Availability of data and materials' section, we presented our data within additional supporting files.

Sincerely,

Lizhi Lv, M.D